Author's response to reviews

Title:A systematic review of the efficacy of self-management programs for increasing physical activity in community-dwelling adults with acquired brain injury (ABI).

Authors:

Taryn M Jones (taryn.jones@mq.edu.au)
Catherine M Dean (catherine.dean@mq.edu.au)
Julia M Hush (julia.hush@mq.edu.au)
Blake F Dear (blake.dear@mq.edu.au)
Nickolai Titov (nick.titov@mq.edu.au)

Version:4 Date:27 March 2015

Author's response to reviews:

Please refer to attached cover letter.